MASSACHUSETTS INSTITUTE OF TECHNOLOGY Department of Physics Physics 8.286 Fall 2009

GRADE HISTOGRAMS

Quiz 1: October 6, 2009

Students: 57 Mean: 64.0 Standard Deviation: 18.5

Bars show bins of 5, arrows show bins of 1.

The grade cuts shown represent my estimate of what you have earned on this quiz, but they will not necessarily be the numerical cuts that will be used to evaluate your final grade. The final grade cuts will be adjusted to account for the level of difficulty of future quizzes and problem sets.